Author's response to reviews

Title: Substance Use and Related Problems Among U.S. Women Who Identify as Mostly Heterosexual

Authors:

Tonda L Hughes (thughes@uic.edu)
Sharon C Wilsnack (sharon.wilsnack@med.und.edu)
Arlinda F Kristjanson (arlinda.kristjanson@med.und.edu)

Version: 2 Date: 13 February 2015

Author's response to reviews:

- Line numbers added
- Updated "Authors Contributions" section